Author's response to reviews

Title: Gossypiboma 19 years after laminectomy mimicking a malignant spinal tumour: a case report

Authors:

Takashi Kobayashi (takakoba825@hotmail.com)
Naohisa Miyakoshi (miyakosh@doc.med.akita-u.ac.jp)
Eiji Abe (abe-e@akikumihsp.com)
Toshiki Abe (toshikiabe@jupiter.ocn.ne.jp)
Tetsuya Suzuki (Akasate@aol.com)
Masato Takahashi (takahashi_masato@akikumihsp.com)
Yoichi Shimada (shimada@hos.akita-u.ac.jp)

Version: 2
Date: 9 August 2014

Author's response to reviews: see over
Dear Editorial Team:

Re: Journal of Medical Case Reports MS: 1628495113127495
“Gossypiboma 19 years after laminectomy mimicking a malignant spinal tumor: a case report”

We would like to thank the reviewers for their time and effort in helping us to improve our manuscript. We are indeed grateful for the opportunity to revise our manuscript in accordance with the issues raised. We have completed the revisions suggested by the reviewers and would like to resubmit our manuscript to Journal of Medical Case Reports.

Following are our point-by-point responses to the comments from the reviewers. We trust that you will find the revisions to your satisfaction and hope you will now consider the manuscript acceptable for publication.

Sincerely,

Takashi Kobayashi, MD, PhD
Department of Orthopedic Surgery
Akita Kosei Medical Venter
1-1-1 Nishifukuro, Iijima, Akita 011-0948, Japan
Tel: +81-18-880-3000; Fax: +81-18-880-3010
E-mail: takakoba825@hotmail.com
Author’s response to reviews
Title: Gossypiboma 19 years after laminectomy mimicking a malignant spinal tumor: a case report

Author: Takashi Kobayashi (takakoba825@hotmail.com)

Version: 2 Date: August 8, 2014

Author's response to reviews:
(1) The histological figures are demonstrative for the readers of the journal. Please add at least one pathologist in the author byline, if possible, because I hope that one contributed to the comment of the histological specimen from Pathology department.

Response: Thank you very much for these valuable comments. We added the pathologist as an author.

(2) Postoperative MRI may be useful to confirm the completeness of the surgical removal of the foreign materials. Please provide postoperative MRI, if possible.

Response: Unfortunately, we do not have postoperative MRI of this patient.

(3) A language revision is necessary before acceptance of the article because there are several misspelling in the text.

Response: We checked the manuscript again. The misspellings were corrected, and they are shown in red and underlined.

(4) Please add arrow to demonstrate the foreign material on the CT and MRI figures (Figs. 1 & 2).
Response: We added open arrows pointing to the foreign material on the CT and MRI figures (Figs. 1 and 2).